Author’s response to reviews

Title: Patent Ductus Arteriosus Endarteritis in a 40-year old woman, diagnosed with Transesophageal Echocardiography.

Authors:

Nikos T Kouris (nikoskou@otenet.gr)
Dr Maria D Sifaki (ngndat@otenet.gr)
Dr Dimitra D Kontogianni (ngndat@otenet.gr)
Ioannis Zaharos (MED4U@winweb.gr)
Dr Eleni M Kalkandi (ngndat@otenet.gr)
Haris E Grassos (harigrass@yahoo.gr)
Dr Dimitris K Babalis (ngndat@otenet.gr)

Version: 2 Date: 7 Mar 2003

Dear Prof Picano
We submit the revision of our manuscript after adding some new data, mainly concerning current invasive management of PDA, according to the reviewers' comments. We also include a movie clip which can replace Figure 3 (figure 2.png), if you judge so.
Yours sincerely
Nikos Kouris MD